Author's response to reviews

Title: In vitro and clinical studies examining the expression of osteopontin in cigarette smoke-exposed endothelial cells and cigarette smokers

Authors:

Emma Bishop (emma_bishop@bat.com)
Eugenia H Theophilus (theophe@rjrt.com)
Ian M Fearon (ian_fearon@bat.com)

Version: 2 Date: 10 January 2012

Author's response to reviews:

As requested we have re-formatted the Abstract and provided information in the Methods section confirming compliance with the Declaration of Helsinki and ethical review of the study. We have also included a statement confirming receipt of informed consent from study subjects.